Author's response to reviews

Title: Delayed school progression and mental health problems in adolescence: a population-based study in 10,803 adolescents.

Authors:

Wanda M. Tempelaar (w.tempelaar@umcutrecht.nl)
Christiaan P. Otjes (chrisotjes@gmail.com)
Clothilde J. Bun (CBun@ggdmn.nl)
Carolien M. Plevier (CPlevier@ggdmn.nl)
Willemijn A. van Gastel (W.vanGastel@umcutrecht.nl)
James H. MacCabe (james.maccabe@kcl.ac.uk)
René S. Kahn (R.Kahn@umcutrecht.nl)
Marco P. Boks (M.P.M.Boks@umcutrecht.nl)

Version: 5 Date: 6 July 2014

Author's response to reviews:

Editorial board
BMC Psychiatry

Utrecht, 5 july 2014

Dear editor,

Thank you for your comments on our manuscript entitled “Delayed school progression and mental health problems in adolescence: a population-based study in 10,803 adolescents”. We are grateful with the opportunity to revise and improve the manuscript.

We have altered the manuscript according to the suggestions:

- we adjusted the manuscript, especially the Discussion, to avoid skipping between tenses.
- We added ‘an’ on page 12.
- We spelled out ‘delayed school progression’ throughout the manuscript.

Yours sincerely,

Wanda Tempelaar, on behalf of all other co-authors

W. Tempelaar
MD, PhDstudent
Dept. of Psychiatry
University Medical Center Utrecht